Author's response to reviews

Title: Comparisons of three anterior cervical surgeries in treating cervical spondylotic myelopathy

Authors:

Ruofu Zhu (zhuruofu@suda.edu.cn)
ZhiDong Wang (sudaguke@163.com)
HuiLin Yang (szyanghuilin@gmail.com)
GenLin Wang (110526039@suda.edu.cn)
MinJie Shen (sudaguke1@163.com)
Quan Yuan (sudaguke2@163.com)

Version: 6 Date: 1 July 2014

Author's response to reviews: see over
Dear Editor:

Thank you for your kind comments for our manuscript to BMC Musculoskeletal Disorders. We appreciate your valuable comments and suggestions to improve it. We have carefully checked our manuscript and also answered the queries point-by-point. The manuscript has been thoroughly checked by a native English speaker. We have modified the various sections so that it conforms to the journal style. We hope that the changes having been made to the manuscript meet to your satisfaction.

Best regards,

Ruofu Zhu, MD, PhD
Department of Orthopaedic Surgery, The First Affiliated Hospital of Soochow University, 188 Shizi Street, Suzhou 215006, China